Author’s response to reviews

Title: Estimating the proportion of Medicaid-eligible pregnant women in Louisiana who do not get abortions when Medicaid does not cover abortion

Authors:
Sarah Roberts (sarah.roberts@ucsf.edu)
Nicole Johns (nicole.e.johns@gmail.com)
Valerie Williams (vwil10@lsuhsc.edu)
Erin Wingo (erin.wingo@ucsf.edu)
Ushma Upadhyay (ushma.upadhyay@ucsf.edu)

Version: 1 Date: 29 May 2019

Author’s response to reviews:

Thank you for pointing out the places where our language in the methods section was too close to our previous publications. We have edited the language to make it different from previous publications and have added more references to our previous publication using this dataset.

We have changed the language in the following lines:

Lines 91 – 95

Lines 99 – 103 (This is the rewrite of the information about abortion funds)

Lines 117-121

Lines 122 and 124

Lines 128, and 131-132

Lines 142-145, 147-149

Lines 158-163